Author's response to reviews

Title: High origin of testicular artery: a case report and review of the literature.

Authors:

George K Paraskevas (g_paraskevas@yahoo.gr)
Orestis Ioannidis (telonakos@hotmail.com)
Athanasios Raikos (a.raikos@yahoo.com)
Basileios Papaziogas (papaziog@med.auth.gr)
Konstantinos Natsis (natsis@med.auth.gr)

Version: 2 Date: 26 May 2010

Author's response to reviews: see over
Dear Editor,

In reply to your email of 24 May 2010, please find attached the updated version of our manuscript MS: 9799724823481883 which has been carefully revised in light of your comments and of those of the three referees.

We hope that this revised version will now be judged ready for publication in JMCY, and looking forward to hearing from you. You can find below a point-by-point answer to the comments.

With warm regards,

Dr. George Paraskevas

Answer to JMCR editor comments:

- The competing interests section after the conclusion was removed as recommended.
- All changes were highlighted with yellow color.
- A structured Abstract was provided according to JMCY style.
- The Ethnicity of the cadaver was added at the Case presentation section.
- A Conclusions section was added.
- Authors contribution was added according to JMCY style.
- A Consent section was updated and corrected according to your remarks.

Answer to referee Alexandros Tzaveas comments:

- The terminology mistake was corrected.
- The cadavers cause of death was added on the Case presentation section.
- TA diameter has been also provided in the Case presentation section.
- Special interest has been provided regarding the clinical condition that may be affected from variant or anomalous TA origin at Discussion section.
- Figure labeling was improved.

Answer to referee Theodoros Moysidis comments:
- More information regarding the clinical significance of the case report has been provided in Discussion and Conclusions section in light of your remarks.

Answer to referee Mohamad Almousa comments:
- The provided Figure is improved and better Figure labeling is available.